Author's response to reviews

Title: Assessment of facility readiness and provider preparedness for dealing with postpartum haemorrhage and pre-eclampsia/eclampsia in public and private health facilities of northern Karnataka, India: a cross-sectional study

Authors:

Krishnamurthy Jayanna Dr (krishnamurthy@khpt.org)
Prem K Mony Dr (premkmony@gmail.com)
Ramesh BM Dr (bmramesh@khpt.org)
Annamma Thomas Dr (rejiann@gmail.com)
Mohan HL Mr (mohan@khpt.org)
James Blanchard Dr (james_blanchard@umanitoba.ca)
Stephen Moses Dr (Stephen.Moses@med.umanitoba.ca)
Lisa Avery Dr (averylisa@hotmail.com)

Version: 2 Date: 23 April 2014

Author's response to reviews:

Dear Editor,

Thanks for your email. I have made changes in terms of adding the page and line numbers.

Warm regards,

Krishnamurthy Jayanna